Author's response to reviews

Title: Duodenal carcinoma at the ligament of Treitz. A molecular and clinical perspective.

Authors:

Peter K Kalogerinis (pkalogerinis@student.methodist.edu)
John E Poulos (poulosj@pol.net)
Andrew Morfesis (Morfesis@owendrivesurgical.com)
Anthony Daniels (Tdani52291@aol.com)
Stavroula Georgakila (stageorga@yahoo.com)
Thomas Daignualt (poulosj@pol.net)
Alexandros G Georgakilas (georgakilasa@ecu.edu)

Version: 4 Date: 13 September 2010

Author's response to reviews:

Dear editors,

We have performed the requested additional changes to our manuscript as following:

1. We have corrected the citations to the appropriate format using the endnote style library provided by the journal. All sources should be correct. There has been no addition or deletion of any sources.

2. We have removed the additional files (figure) from the manuscript, as you already have them in your records, and have it available for publication purposes. We have also removed any blank pages from the document. We have left the Table still included in the manuscript.

3. We have corrected the authors’ contribution section as you requested. It remains after the conclusion with the acknowledgements and disclaimer.

4. We have adjusted the title of the manuscript, and corrected the format for author's names.

5. We have put the abbreviations before the references, and have formatted them as a paragraph.

Thank you for your acceptance of our manuscript.

Alexandros Georgakilas, Ph.D
Associate Professor
Head of DNA Damage and Repair Laboratory
Biology Department
Howell Science Complex
East Carolina University